LSD Deputy Report October 11 – October 24, 2012

Mary Logue

Highlights:

	Completed	Continued
Cryo		Transfer line PMInstall He line to Hall D
FM&L	Hall C Truck Ramp Penetration Dominion 40 MVA station PM	 CHL Cooling Tower piping installation
Halls		De-installation work
I&C		 Magnet communication lines
INST	 Lots of stands, girders, base and ceiling plates WArc PSS gate installed 4 YA magnets moved to MMF 2 BComm dipole magnets removed East Arc lintel 	Hall D lintel installation

Highlights

	Completed	Continued
Power PM	Emergency Power Panel	House Power
PSS		 Installing cables for extension to Hall D – lots of cable pulls
RF		C-100 RF upstairsTurn on to filaments
S&A	 1S stand alignment OR/OL girder stand alignment SL03 alignment SLinac alignment completed East Arc 2A alignment Line D layout East Recombiner fiducialization 	• East Arc 2A
SRF		PM ongoing



